Author’s response to reviews

Title: Evaluation of Tp-e Interval, Tp-e/QT Ratio and Tp-e/QTc Ratio in Patients with Acute Myocarditis

Authors:
Fatih Mehmet Uçar (dr_fmucar@hotmail.com)
Cihan OZTURK (dr.cihanozturk@gmail.com)
Mustafa Adem YILMAZTEPE (mayilmaztepe@yahoo.com)

Version: 4 Date: 27 Aug 2019

Author’s response to reviews:

Editor Comments:
1. It has come to our attention that within your manuscript there is significant text overlap with other publications, particularly:

Introduction:

Materials and Methods:

Discussion:

Conclusion:

RESPONSE: Thank you so much for your evaluation. We revised these parts as your comments. We used iThenticate: Plagiarism Detection Software for detailed assessment. Please re-evaluate the manuscript. Best regards.